Author’s response to reviews

Title: The risk of AIDS-defining events is decreasing over time in the German HIV-1 Seroconverter Cohort

Authors:

Mathias Altmann (altmannm@rki.de)
Matthias an der Heiden (anderheidenm@rki.de)
Ramona Scheufele (scheufeler@rki.de)
Katrin Hartmann (katiliba@aol.com)
Claudia Houareau (claudiahouareau@web.de)
Barabara Bartmeyer (Gunsenheimer-BartmeyerB@rki.de)
Osamah Hamouda (hamoudao@rki.de)

Version: 3 Date: 14 September 2011

Author’s response to reviews:

Dear Mrs Roselle Pangilinan,

As you asked in your Email, I have made the following changes:
- list of Email adresse of the authors (title page)
- make the author´s affiliation clearer (title page)
- delete the running head (title page)
- add a sentence in the background of the abstract: "With ageing of the HIV-infected population, long-term exposure to treatment, varying adherence, emerging resistance and complications to therapies, (..)" (abstract page)

I hope these changes will meet your requests in order to start the peer review process.

Thank you very much for your collaboration in this work.

Mathias Altmann